Author’s response to reviews

Title: Never again? Challenges in transforming the health workforce landscape in post-Ebola West Africa

Authors:

Barbara McPake (bmcpake@unimelb.edu.au)

Prarthna Dayal (pdayal@unimelb.edu.au)

Christopher Herbst (cherbst@worldbank.org)

Version: 2 Date: 10 Feb 2019

Author’s response to reviews:

Many thanks for the careful review of our resubmission. Please find responses in the table provided in the cover letter. (It seems that the web interface does not allow the pasting of a table in this section.)